Author's response to reviews

Title: The development of a brief and objective method for evaluating moral sensitivity and reasoning in medical students.

Authors:

Prof Akira Akabayashi (akirasan-tky@umin.ac.jp)
Brian T Slingsby (taylor@pbh.med.kyoto-u.ac.jp)
Ichiro Kai (ichirokai-tky@umin.ac.jp)
Prof Tadashi Nishimura (akirasan-tky@umin.ac.jp)
Akiko Yamagishi (yamagishi@nurs.juntendo.ac.jp)

Version: 4 Date: 29 Jan 2004

Editor: We have revised as you suggested. Figures are cropped, X axis title is completed, minor changes in wording have been done. We hope it is now acceptable for publication.